# PLOS MEDICINE



# GOPEN ACCESS

**Citation:** O'Brien KS, Arzika AM, Maliki R, Manzo F, Mamkara AK, Lebas E, et al. (2020) Biannual azithromycin distribution and child mortality among malnourished children: A subgroup analysis of the MORDOR cluster-randomized trial in Niger. PLoS Med 17(9): e1003285. https://doi. org/10.1371/journal.pmed.1003285

Academic Editor: Lars Åke Persson, London School of Hygiene and Tropical Medicine, UNITED KINGDOM

Received: April 13, 2020

Accepted: August 10, 2020

Published: September 15, 2020

**Copyright:** © 2020 O'Brien et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** The de-identified, individual-level data from the trial are publicly available through the Open Science Framework: https://osf.io/hxg59/.

**Funding:** TML received the Bill and Melinda Gates Foundation grant (OP1032340, <u>https://www.</u> <u>gatesfoundation.org/</u>) for the MORDOR trial. CEO received a grant from Research to Prevent Blindness (no number, <u>https://ophthalmology.ucsf.</u> RESEARCH ARTICLE

# Biannual azithromycin distribution and child mortality among malnourished children: A subgroup analysis of the MORDOR clusterrandomized trial in Niger

Kieran S. O'Brien<sup>1,2</sup>\*, Ahmed M. Arzika<sup>3</sup>, Ramatou Maliki<sup>3</sup>, Farouk Manzo<sup>3</sup>, Alio K. Mamkara<sup>3</sup>, Elodie Lebas<sup>1</sup>, Catherine Cook<sup>1</sup>, Robin L. Bailey<sup>4</sup>, Sheila K. West<sup>5</sup>, Catherine E. Oldenburg<sup>1</sup>, Travis C. Porco<sup>1,6,7,8</sup>, Benjamin Arnold<sup>1,6</sup>, Jeremy D. Keenan<sup>1,6</sup>, Thomas M. Lietman<sup>1,6,7,8</sup>, for the MORDOR Study Group

Francis I. Proctor Foundation, University of California, San Francisco, California, United States of America,
Division of Epidemiology, School of Public Health, University of California, Berkeley, California, United States of America,
The Carter Center, Niamey, Niger,
Clinical Research Unit, Department of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom,
Dana Center for Preventive Ophthalmology, Wilmer Eye Institute, Johns Hopkins University, Baltimore, Maryland, United States of America,
Department of Ophthalmology, University of California, San Francisco, California, United States of America,
Department of Epidemiology and Biostatistics, University of California, San Francisco, California, United States of America,
Institute for Global Health Sciences, University of California, San Francisco, California, United States of America,

\* kieran.obrien@ucsf.edu

# Abstract

# Background

Biannual azithromycin distribution has been shown to reduce child mortality as well as increase antimicrobial resistance. Targeting distributions to vulnerable subgroups such as malnourished children is one approach to reaching those at the highest risk of mortality while limiting selection for resistance. The objective of this analysis was to assess whether the effect of azithromycin on mortality differs by nutritional status.

# Methods and findings

A large simple trial randomized communities in Niger to receive biannual distributions of azithromycin or placebo to children 1–59 months old over a 2-year timeframe. In exploratory subgroup analyses, the effect of azithromycin distribution on child mortality was assessed for underweight subgroups using weight-for-age Z-score (WAZ) thresholds of -2 and -3. Modification of the effect of azithromycin on mortality by underweight status was examined on the additive and multiplicative scale. Between December 2014 and August 2017, 27,222 children 1–11 months of age from 593 communities had weight measured at their first study visit. Overall, the average age among included children was 4.7 months (interquartile range [IQR] 3–6), 49.5% were female, 23% had a WAZ < -2, and 10% had a WAZ < -3. This analysis included 523 deaths in communities assigned to azithromycin and 661 deaths in communities assigned to placebo. The mortality rate was lower in communities assigned to azithromycin than placebo overall, with larger reductions among children with lower WAZ: edu/research-prevent-blindness-awards/). Pfizer provided the medication used in this study, Neither Pfizer nor the funders had any role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

**Abbreviations:** GEE, generalized estimating equation; IQR, interquartile range; MORDOR, Macrolides Oraux pour Réduire les Décès avec un Oeil sur la Résistance; RERI<sub>HR</sub>, Relative Excess Risk due to Interaction from hazard ratios; WAZ, weight-for-age Z-score. -12.6 deaths per 1,000 person-years (95% CI -18.5 to -6.9, P < 0.001) overall, -17.0 (95% CI -28.0 to -7.0, P = 0.001) among children with WAZ < -2, and -25.6 (95% CI -42.6 to -9.6, P = 0.003) among children with WAZ < -3. No statistically significant evidence of effect modification was demonstrated by WAZ subgroup on either the additive or multiplicative scale (WAZ < -2, additive: 95% CI -6.4 to 16.8, P = 0.34; WAZ < -2, multiplicative: 95% CI 0.8 to 1.4, P = 0.50, WAZ < -3, additive: 95% CI -2.2 to 31.1, P = 0.14; WAZ < -3, multiplicative: 95% CI 0.9 to 1.7, P = 0.26). The estimated number of deaths averted with azithromycin was 388 (95% CI 214 to 574) overall, 116 (95% CI 48 to 192) among children with WAZ < -2, and 76 (95% CI 27 to 127) among children with WAZ < -3. Limitations include the availability of a single weight measurement on only the youngest children and the lack of power to detect small effect sizes with this rare outcome. Despite the trial's large size, formal tests for effect modification did not reach statistical significance at the 95% confidence level.

# Conclusions

Although mortality rates were higher in the underweight subgroups, this study was unable to demonstrate that nutritional status modified the effect of biannual azithromycin distribution on mortality. Even if the effect were greater among underweight children, a nontargeted intervention would result in the greatest absolute number of deaths averted.

# **Trial registration**

The MORDOR trial is registered at clinicaltrials.gov NCT02047981.

# Author summary

# Why was this study done?

- Previous research has found that distributing oral azithromycin to children 1–59 months old biannually reduces mortality in that age group.
- Despite the promise of this intervention to promote child survival in high-mortality settings, community distribution of antibiotics like azithromycin has the potential to increase antimicrobial resistance, which could reduce the efficacy of azithromycin and similar antibiotics over time.
- To harness the benefit of the intervention while reducing the potential for increasing resistance, some propose targeting the intervention to smaller groups of the population that face a particularly high risk of mortality.
- As malnutrition is implicated in a large proportion of childhood mortality, malnourished children are one potential subgroup to target.

# What did the researchers do and find?

• We conducted a subgroup analysis using data from a trial that compared communities randomized to azithromycin distribution or placebo distribution to children 1–59 months over 2 years in Niger.

- We included 27,222 children from the original trial who had weight measured at their first study visit and compared the effect of azithromycin to placebo on mortality among those who were not underweight, were underweight, and were severely underweight, and we calculated the number of deaths that would have been averted if this intervention had been given to the overall group as well as these groups defined by underweight status.
- The observed effect of azithromycin on mortality was strongest in the severely underweight subgroup, but we found no statistical evidence that the effect differed by underweight status.
- As only 10% of children were in the severe underweight group, the overall number of deaths averted would be greatest with an intervention treating all children 1–11 months, rather than only underweight children.

#### What do these findings mean?

- The mortality rates and the observed effect of azithromycin on mortality were largest for the severe underweight group, suggesting a potential group to target with this intervention, although the study was not powered to detect this effect.
- Even if this study provided evidence that the effect was strongest among underweight children, a nontargeted intervention would have prevented the greatest number of deaths in this population.

# Introduction

Biannual azithromycin distribution reduced mortality among children 1–59 months of age in a large cluster-randomized trial in Malawi, Niger, and Tanzania (MORDOR trial, "Macrolides Oraux pour Réduire les Décès avec un Oeil sur la Résistance") [1,2]. The strongest effects were observed in Niger, which had the highest baseline mortality rates, and in children 1–11 months of age [1]. In conjunction with existing child survival activities, this intervention has the potential to bolster progress in reducing under-5 mortality, particularly in high-mortality settings. However, these distributions increase the prevalence of antimicrobial resistance [3,4]. Limiting antibiotic distributions to smaller subgroups at the highest risk of mortality might be an approach to reduce selection for resistance [5].

Malnutrition is implicated in up to 45% of all childhood deaths globally [6]. Malnourished children are at increased risk of mortality from infectious diseases such as diarrhea and respiratory tract infections [6]. Moreover, the relationship between malnutrition and infection is complex, with undernutrition suppressing the immune system and increasing the risk of infection, and infection causing a reduction in appetite, malabsorption of nutrients, and competition for nutrients [7,8], Provision of antibiotics to malnourished children could lead to clearance of both overt and subclinical infections associated with mortality. Use of antibiotics with a long half-life, like azithromycin, could also prevent the development of infections during the 1–2 weeks after administration [9]. Other proposed mechanisms for a beneficial effect of antibiotics in undernourished children involve modulation of the intestinal microbiota,

which could result in a reduction in gut flora that compete for nutrients and affect chronic conditions like environmental enteropathy [8,10–14].

Multiple studies have examined the role of antibiotics in malnourished children, with varying results. Three individual-randomized trials have compared antibiotics to placebo in the management of severe acute malnutrition [12,15,16]. One trial in Malawi found that children receiving antibiotics experienced greater nutritional recovery and less mortality than those receiving placebo [15], whereas two other trials found no difference in either nutritional recovery or mortality between arms [12,16]. Fewer studies have focused on children with moderate malnutrition, although one multi-country trial evaluating the effect of antibiotics on a number of outcomes in children with moderate acute malnutrition is currently underway [17].

Targeting high-risk subgroups such as malnourished children with azithromycin could preserve resources and lower the risk of selecting for antimicrobial resistance. However, evidence on the effect of antibiotics on mortality in malnourished children is mixed. The MORDOR trial provides an opportunity to examine the role of antibiotics in reducing mortality in malnourished children in a sub-Saharan African setting. The objective of this subgroup analysis was to assess whether the effect of biannual distribution of oral azithromycin on child mortality differed by nutritional status in Niger.

#### Methods

#### Trial design, setting, and participants

MORDOR was a large, simple, multisite cluster-randomized trial designed to compare the effect of biannual distribution of oral azithromycin to placebo on child mortality [1]. The protocol and statistical analysis plan for the main trial have been published, and the analyses presented here are exploratory [1]. This analysis included the Niger site, which enrolled communities in the Boboye and Loga districts (now Boboye, Loga, and Falmey districts after nation-wide redistricting). Communities with populations between 200 and 2,000 inhabitants according to the Niger 2012 census were eligible for inclusion in the main trial. Children 1–59 months of age who weighed  $\geq 3.8$  kg were eligible for treatment. This subgroup analysis included children 1–11 months old who had weight recorded at the time of the child's first census, which could have been in any one of the censuses. Children 12–59 months old were excluded because crude height intervals were used to determine dose in children able to stand, and nutritional status indicators could not be accurately calculated for this group.

Ethical approval for the Niger site was obtained from the Niger Ministry of Health and the University of California, San Francisco Committee on Human Research. Verbal informed consent was obtained from households and caregivers before inclusion. The trial was conducted in accordance with the principles of the Declaration of Helsinki and was registered at Clinicaltrials.Gov (NCT02047981).

#### Census

A door-to-door census was conducted every 6 months to enumerate households in the study area between December 2014 and August 2017. Demographic information (age, sex) was recorded for each child 1–59 months old. During follow-up census data collection, vital status (alive, dead, or unknown) and residence (living in community, moved outside community, or unknown) were recorded. Five censuses (four inter-census periods) were completed during the 2-year study. Data were collected electronically using a custom-designed mobile application (Conexus, Los Gatos, CA) and uploaded to a cloud-based server (Salesforce, San Francisco, CA).

#### Interventions

At every biannual census, each child 1–59 months old was offered a single, directly observed dose of oral azithromycin or placebo (Pfizer, New York, NY). Children were given a dose of 20 mg per kg, which was assessed by height-stick approximation according to Niger's trachoma program guidelines or by weight for children unable to stand. Children known to be allergic to macrolides were not treated. Adverse events were monitored and have been reported elsewhere [1,18].

#### Outcomes

The outcome for this analysis is mortality, defined as community mortality rate (deaths per 1,000 person-years at risk). Data collected during the biannual census were used to assess the outcome. A death was included if a child was recorded as alive on one census and died at the subsequent census. Person-time at risk was calculated as the number of days between consecutive census periods or until death. Children who moved or had an unknown status at the subsequent census contributed half of the days during that inter-census period.

# Assessment of nutritional status

The trial protocol included assessment of weight for the purpose of determining dosage in children unable to stand. Trained study personnel recorded weight (if measured) and dose administered for all children in the mobile application. To determine dosage, children unable to stand were weighed (Amw-tl440 digital hanging scale, American Weigh Scales, Cumming, GA), and weight was recorded to the nearest 0.1 kg. A single weight measurement was taken at each visit. Age- and sex-adjusted weight-for-age Z-scores (WAZs) were calculated using the 2006 WHO Child Growth Standards with the zscorer package in R (R Foundation for Statistical Computing, Vienna, Austria) [19–21]. WAZ was dichotomized to group children without or with moderate to severe malnutrition (WAZ  $\geq$  –2 and WAZ < –2, respectively) and without or with severe malnutrition (WAZ  $\geq$  –3 and WAZ < –3, respectively). These categories were chosen to align with current classification standards used in nutritional policies and programs. Children with a baseline WAZ of less than –6 or greater than 5 were excluded according to WHO recommendations [20]. As WAZ was calculated after program completion, underweight children were not actively identified during the study period, and no additional measures were taken to address nutritional status during the trial.

# Randomization and masking

Within each country, communities were randomized 1:1 to receive biannual azithromycin or placebo. The randomization sequence was generated in R by the trial biostatistician and was implemented by unmasked members of the data team and Pfizer. The allocation was concealed by simultaneous randomization assignment. Participants, investigators, data collectors, and data analysts were masked to treatment assignment. Placebo was packaged to be identical in appearance to the azithromycin to maintain masking.

#### Sample size and statistical methods

The MORDOR trial was designed and powered for the primary outcome, which has been previously published [1]. Briefly, the overall trial had 80% power to detect a 10% difference in allcause mortality among communities receiving azithromycin compared to placebo, and the Niger site included 594 eligible communities [1]. Given the fixed design, the prevalence of moderate to severe and severe underweight, and the mortality rates within subgroups, this subgroup analysis had 80% power to detect additive interaction effects of the following sizes, interpreted as the mortality rate among underweight children receiving placebo in excess of the individual effects of underweight or placebo on mortality: 17 deaths per 1,000 person-years for the moderate to severe subgroup and 25 deaths per 1,000 person-years for the severe subgroup [22].

Analyses were conducted in R. Participant characteristics, WAZ, and outcomes were summarized by arm using frequency and percentage for categorical variables, mean and standard deviation for continuous variables, and incidence rate (deaths per 1,000 person-years, hereafter referred to as "mortality rate") and 95% confidence interval for outcomes. Confidence intervals were constructed using percentiles from bootstrap resampling with 1,000 replicates. Participant characteristics were also compared among those included in the analysis and those excluded for having missing or invalid weight measurements. No multiple comparisons corrections were made.

Effect modification was evaluated non-parametrically with interaction contrasts [23]. To calculate the contrasts, subgroups were coded such that the groups with the lowest mortality rates were the reference categories (i.e.,  $R_{00}$  = mortality rate among higher-weight children in communities assigned to azithromycin,  $R_{10}$  = mortality rate among underweight children in communities assigned to placebo, and  $R_{11}$  = mortality rate among underweight children in communities assigned to placebo [24]. An additive interaction contrast greater than 0 indicates the joint effect of receiving placebo and being underweight is greater than the sum of the individual effects considered separately. A multiplicative interaction contrast greater than 1 indicates the joint effect of receiving placebo and being underweight is greater than the product of the individual effects considered separately. The absolute number of deaths averted with azithromycin in each subgroup was also estimated using person-time at risk in both arms and the subgroup-level mortality rates.

Several sensitivity analyses were conducted. Survival probability was summarized by treatment arm and WAZ subgroup using Kaplan-Meier survival curves. Effect modification was also examined using Cox proportional hazards models. To determine the presence of multiplicative interaction, models included a shared frailty assuming a gamma distribution to account for clustering, the Efron method for ties, and treatment and WAZ as covariates with their product as an interaction term. Model estimates were reported with hazard ratios for each subgroup against a single reference category and with hazard ratios for the effect of treatment within each stratum of WAZ [23,25]. The estimated hazard ratios were used to calculate the Relative Excess Risk due to Interaction (RERI<sub>HR</sub>) to assess the presence and direction of additive interaction, with the same coding as used for the interaction contrasts [23-26]. The delta method was used to calculate standard errors for the RERIHR [23]. As treatment arm was randomized and is the primary intervention of interest, confounding of the relationship between nutritional status and mortality was not considered, and no additional factors were controlled for in the models [23]. Model assumptions were evaluated graphically with ln(-ln) survival plots and analytically with tests of scaled Schoenfeld residuals as well as with models including terms for interactions with time to event for each covariate. The appropriateness of the distributional assumptions for the shared frailty were assessed by comparing results against models using a lognormal distribution for the shared frailty and estimated with generalized estimating equations (GEEs) to account for clustering.

Additional sensitivity analyses included evaluating the potential for bias introduced by the selection of the analysis sample by restricting the analysis to children eligible during the first inter-census period only and by restricting to children 1–5 months of age. To assess the impact of the use and form of WAZ, baseline weight, age, and sex were included in the models, and baseline WAZ was assessed in continuous form. To evaluate assumptions made in determining

time to mortality when no exact date was available, an interval censoring method was also used. This was implemented as a generalized linear mixed model, with a binary outcome for death, a complementary log-log link, a term for inter-census period, and a random effect for community.

#### Results

In December 2014, 615 communities in Niger were randomized to receive biannual azithromycin or placebo in the main trial, of which 594 communities were successfully censused and included in analyses (Fig 1). Treatment coverage among children 1–59 months old was greater than 91% over the 4 inter-census periods in both arms. The final sample for this analysis included 593 communities with 27,222 children 1–11 months old who had a valid weight recorded at the time of the child's first entry into the study. One community was not included because it had no eligible children, and 12,086 children 1–11 months old at their first census were excluded either for having no weight recorded (11,899 children, of which 10,271 had approximate height measured) or having a WAZ less than –6 or greater than 5 recorded (187 children). Over the 2-year study period, 5,189 children were lost to follow-up, with a similar percentage of children lost in each arm.

Characteristics of included children at the time of the child's first census are shown by treatment arm in Table 1. Overall, the median age was 4 months (interquartile range [IQR] 3–6), and 49.5% of children (13,484/27,222) were female. Mean WAZ was -0.8 (SD 1.7), with 23.0% (6,268/27,222) of all children having a WAZ < -2 and 10.1% (2,755/27,222) having a WAZ < -3. All characteristics were similar in both arms. Excluded children were older than included children (median age 9 months, IQR 6–11), and a similar percentage were female (49.1%; S1 Table).

The analysis included 1,184 deaths and a total of 30,852 person-years at risk (Table 2). The overall difference in the incidence of mortality comparing communities assigned to azithromycin to communities assigned to placebo was -12.6 deaths per 1,000 person-years (95% CI -18.5 to -6.9, P < 0.001). By subgroup, this difference was -17.0 (95% CI -28.0 to -7.0, P = 0.001) among those with WAZ < -2, and -25.6 (95% CI -42.6 to -9.6, P = 0.003) among those with WAZ < -3. Fig 2 compares mortality rates by treatment arm and subgroup. Interaction contrasts on the additive scale were 5.7 deaths per 1,000 person-years (95% CI -6.4 to 16.8, P = 0.34) for the moderate to severe subgroup and 14.4 deaths per 1,000 person-years (95% CI -2.2 to 31.1, P = 0.14) for the severe subgroup. On the multiplicative scale, these contrasts were 1.1 (95% CI 0.8 to 1.4, P = 0.50) and 1.2 (95% CI 0.9 to 1.7, P = 0.26), respectively. The estimated number of deaths averted with azithromycin among children 1–11 months old was 388 (95% CI 214 to 574) overall, 116 (95% CI 48 to 192) among children with WAZ < -2, and 76 (95% CI 27 to 127) among children with WAZ < -3.

Figs 3 and 4 display survival probabilities by arm and subgroup, and Table 3 reports modelbased estimates of mortality and effect modification by subgroup. Among children in placebotreated communities, lower WAZ was associated with an increased hazard of mortality (HR 1.32, 95% CI 1.11–1.57, P = 0.002 comparing WAZ < -2 to WAZ  $\geq$  -2 and HR 1.56, 95% CI 1.25–1.95, P < 0.001 comparing WAZ < -3 to WAZ  $\geq$  -3). The hazard for mortality was lower in communities assigned to azithromycin than communities assigned to placebo, with a more pronounced effect for the subgroups of underweight children (27% lower in WAZ  $\geq$  -2, 95% CI 15–38, P < 0.001; 30% lower in WAZ < -2, 95% CI 11–45, P = 0.003; and 38% lower in WAZ < -3, 95% CI 14–55, P = 0.005). When comparing underweight children in communities assigned to azithromycin to higher-weight children in communities assigned to placebo, the hazards for mortality were similar in both subgroups (HR 0.93, 95% CI 0.75–1.14, P = 0.48





Characteristic		Azithromycin <i>n</i> = 14,243	Placebo n = 12,979
Age, months, median (IQR)		4 (3-7)	4 (3-6)
Female sex, n (%)		7,040 (49.4%)	6,444 (49.6%)
Census period of entry into study, n (%)			
	1	4,470 (31.1%)	4,034 (31.4%)
	2	3,880 (28.3%)	3,673 (27.2%)
	3	2,751 (20.0%)	2,592 (19.3%)
	4	3,142 (20.6%)	2,680 (22.1%)
WAZ, mean (SD)		-0.8 (1.7)	-0.8 (1.7)
WAZ category, moderate to severe, $n$ (%)			
	$\geq -2$	10,988 (77.1%)	9,966 (76.8%)
	< -2	3,255 (22.9%)	3,013 (23.2%)
WAZ category, severe, n (%)			
	$\geq -3$	12,796 (89.8%)	11,671 (89.9%)
	< -3	1,447 (10.2%)	1,308 (10.1%)

Table 1. Characteristics of children 1-11 months old with weight recorded at the time of entry into the study.

Abbreviations: IQR, interquartile range; WAZ, weight-for-age Z-score

https://doi.org/10.1371/journal.pmed.1003285.t001

comparing WAZ < -2 to WAZ  $\ge -2$  and HR 0.97, 95% CI 0.74–1.27, P = 0.82 comparing WAZ < -3 to WAZ  $\ge -3$ ). No evidence of effect modification was identified. Similar results were found in all sensitivity analyses.

#### Discussion

This subgroup analysis evaluated whether the effect of biannual azithromycin distribution on child mortality differed by underweight status in a high-mortality West African setting. Azithromycin was associated with an overall 28% reduction in mortality compared to placebo in children 1–11 months old with weight measured, similar to the age-based subgroup results from the main trial [1]. As expected given evidence on the relationship between malnutrition and mortality [6,27,28], lower weight for age was associated with increased mortality. The observed time to mortality in underweight children receiving azithromycin was approximately the same as that of higher-weight children receiving placebo. Although the absolute reduction in mortality between arms appears larger in both underweight groups, no evidence of effect modification by WAZ subgroup was found at the 95% confidence level. The number of deaths averted was greatest if all children were treated with azithromycin, regardless of nutritional status.

The nonspecific distribution of azithromycin to reduce child mortality presents an ethical dilemma: given the strong evidence of efficacy, it may be unethical to withhold such an intervention, yet the intervention's effect on antimicrobial resistance warrants caution [29]. Increasing resistance could reduce the efficacy of essential antibiotics, potentially causing additional morbidity and mortality in the longer term. Targeting the intervention to high-risk subgroups is one solution to preserve resources and reduce negative consequences; targeting all children 1–11 months in this study population required 10 times the amount of azithromycin compared to targeting WAZ < -3. A targeted approach may also be more cost-effective than a broader distribution strategy [30]. The assumption that targeting vulnerable subgroups results in the greatest population health benefits has been questioned, however, since more lives are saved by intervening on a population with a wider risk spectrum [31–33]. Here, although there

Category	Azithromycin				Placebo				Mortality rate	Mortality rate
	n	Deaths	Person-years at risk	Mortality rate <sup>1</sup> (95% CI) <sup>2</sup>	n	Deaths	Person-years at risk	Mortality rate <sup>1</sup> (95% CI) <sup>2</sup>	ratio (95% CI) <sup>2</sup>	difference (95% CI) <sup>2</sup>
Overall	14,243	523	16,153	32.4 (29.3 to 35.5)	12,979	661	14,699	45.0 (40.3 to 49.7)	0.72 (0.62 to 0.83)	-12.6 (-18.5 to -6.9)
WAZ category, moderate to severe										
≥-2	10,988	387	12,610	30.7 (27.0 to 34.4)	9,966	480	11,435	42.0 (36.9 to 47.3)	0.73 (0.61 to 0.87)	-11.3 (-17.5 to -5.2)
< -2	3,255	136	3,543	38.4 (32.4 to 44.9)	3,013	181	3,264	55.4 (46.7 to 64.9)	0.69 (0.54 to 0.86)	-17.0 (-28.0 to -7.0)
WAZ category, severe										
≥-3	12,796	460	14,599	31.5 (28.3 to 35.0)	11,671	568	13,293	42.7 (37.6 to 47.6)	0.74 (0.63 to 0.84)	-11.2 (-17.6 to -5.3)
<-3	1,447	63	1,554	40.5 (30.9 to 49.7)	1,308	93	1,406	66.1 (53.7 to 79.4)	0.61 (0.44 to 0.84)	-25.6 (-42.6 to -9.6)

Table 2. Number of deaths	, person-time at risk, and mortalit	y rates by treatment arm an	d subgroups of WAZ.
---------------------------	-------------------------------------	-----------------------------	---------------------

<sup>1</sup>Deaths per 1,000 person-years at risk.

<sup>2</sup>Confidence intervals calculated using bootstrap resampling at the community level to account for clustering; 1,000 replicates were used.

Abbreviation: WAZ, weight-for-age Z-score

https://doi.org/10.1371/journal.pmed.1003285.t002

is some indication that intervening on those with the lowest WAZ may be particularly beneficial, the absolute number of deaths averted was 5 times greater when including all children 1–11 months as opposed to only the 10% with WAZ < -3. In addition, possible indirect effects might be lost with a more focused intervention. Finally, targeting a subgroup of the population presents its own ethical complexity, as providing a beneficial intervention more broadly might be more equitable when resources are available to do so [29].

Approximately 23% of the children included in this analysis were underweight, similar to other estimates indicating that Niger bears a high burden of malnutrition [34]. A single weight measurement was taken on a subset of children 1–11 months old who were unable to stand, which has several implications for interpretation of these results. First, other nutritional status indicators like wasting and stunting were not assessed, nor were the causes of underweight status. Underweight status has been shown to increase the risk of mortality in multiple settings [6,27,28,35,36], with some evidence demonstrating that WAZ alone is a highly sensitive and specific indicator of concurrent wasting and stunting [37]. In addition, as malnutrition is caused by a wide variety of factors, azithromycin might be more effective in cause-specific subgroups of underweight children, though this study was neither designed nor powered to assess the effect by smaller subgroups. Similarly, as underweight status could be a proxy for other child, household, and community characteristics, the mechanism of effect modification is likely more complex than modeled here. Second, although being underweight at the first visit likely predicts being underweight at later visits, we were unable to unable to examine the impact of changing nutritional status over time. Children who became underweight after their first visit thus might be misclassified by this analysis, which we would expect to bias any effect modification towards the null. Third, the selection of children 1-11 months of age who had weight measurements available could introduce bias, since children at the older end of that range who were able to stand were more likely not to be weighed. However, exclusions among the older age group were balanced by arm, overall and across census periods, and sensitivity analyses restricting the population to children 1-5 months produced similar results to the



**Fig 2. Comparison of mortality rates by treatment arm and WAZ subgroup with interaction contrasts.** (A, B) Comparisons of mortality rate (deaths per 1,000 person-years) by treatment arm overall and by WAZ subgroup on the additive (A) and multiplicative (B) scales. (A) Mortality rate differences (mortality rate in communities assigned to placebo). (B) Mortality rate ratios (mortality rate in communities assigned to placebo). (C, D) Interaction contrasts on the additive (C) and multiplicative (D) scales. Interaction contrasts defined subgroups such that the groups with the lowest mortality rates were the reference categories (i.e.,  $R_{00}$  = mortality rate among higher-weight children in communities assigned to placebo), and  $R_{11}$  = mortality rate among underweight children in communities assigned to placebo). (C) Interaction contrasts on the additive scale. (D) Interaction contrasts on the multiplicative scale. WAZ, weight-for-age Z-score.

https://doi.org/10.1371/journal.pmed.1003285.g002

main analysis. Additionally, older children were not weighed. The analysis population thus might not be representative of the general population, as it might include a higher prevalence of underweight children and does not reflect the experience of children 12–59 months old. Fourth, the SD for WAZ was greater than 1 [20], likely due to measurement error since weight was assessed primarily for the purpose of intervention delivery. Only one measurement was taken at each visit for each child in order to determine dosage. As mean WAZ and SD were similar across arms, any information bias is likely to be conservative, which could have masked the presence of effect modification. Fifth, both mortality and malnutrition are known to vary seasonally in West Africa [38,39]. Seasonality-focused analyses were not pursued given the low



Fig 3. Kaplan-Meier estimates of survival probability by treatment arm and the moderate to severe underweight subgroup. Each curve depicts a different subgroup, with placebo represented by dotted lines in shades of blue and azithromycin represented by solid lines in shades of red. The darker shades indicate the higher-weight subgroup (WAZ  $\geq -2$ ), and the lighter shades indicate the underweight subgroup (WAZ < -2). The y-axis is broken for clarity and jumps from 0.00 to 0.85. WAZ, weight-for-age Z-score.

https://doi.org/10.1371/journal.pmed.1003285.g003

power to further stratify the population, and the lack of an overall seasonal effect of azithromycin on mortality in the main trial [39]. Finally, the use of cutoffs to categorize malnourished groups has been criticized for creating a false separation of subgroups in which to intervene [40], particularly in high-burden areas where the entire distribution of anthropometric indicators is shifted downwards. As these cutoffs are actively used in current programs and policy, their use in this application provides readily available information to these sectors while also calling into question the impact of a targeted strategy that would exclude many children with mild to moderate malnutrition who also face an increased burden of mortality [27].

Additional limitations of this study include those shared by most subgroup analyses of trials, such as the potential for false negatives from lack of power and bias from use of improper subgroups. The effect sizes observed in this analysis were smaller than detectable by the design (5.7 versus 17 deaths per 1,000 person-years for the moderate to severe subgroup, and 14.4 versus 25 deaths per 1,000 person-years for the severe subgroup), indicating the analysis was



Fig 4. Kaplan-Meier estimates of survival probability by treatment arm and the severe underweight subgroup. Each curve depicts a different subgroup, with placebo represented by dotted lines in shades of blue and azithromycin represented by solid lines in shades of red. The darker shades indicate the higher-weight subgroup (WAZ  $\geq -3$ ), and the lighter shades indicate the underweight subgroup (WAZ < -3). The y-axis is broken for clarity and jumps from 0.00 to 0.85. WAZ, weight-for-age Z-score.

https://doi.org/10.1371/journal.pmed.1003285.g004

underpowered. The use of baseline WAZ from children who entered the study after azithromycin had been distributed at the community level could result in bias since WAZ for these children is a post-randomization characteristic that could be influenced by treatment arm. A sensitivity analysis restricted to the first phase of the study did not reveal differences in results. Also, underweight prevalence did not differ by arm across census period, so more complex approaches to assessing or controlling for this potential bias were not pursued. In this type of dynamic cohort, differential loss to follow-up can result in selection bias. Although loss to follow-up was present, it was similar when compared by arm. Further research would be required to determine whether these results were generalizable to settings beyond those similar to Niger, which has a high burden of both malnutrition and mortality. Strengths of this study include the large sample size, the assessment of both additive and multiplicative interaction, and the randomized design.

WAZ category		Hazard ratio	s (95% CI)	Measures of effect modification (95% CI)		
	Placebo	Azithromycin	Azithromycin within strata of WAZ	RERI <sub>HR</sub> (additive) <sup>2</sup>	Ratio of HRs (multiplicative)	
Moderate to severe						
$\geq -2$	1 (ref)	0.73 (0.62 to 0.85)	0.73 (0.62 to 0.85)	0.17 (-0.20 to 0.55)	0.96 (0.74 to 1.25)	
< -2	1.32 (1.11 to 1.57)	0.93 (0.75 to 1.14)	0.70 (0.55 to 0.89)			
Severe						
≥-3	1 (ref)	0.74 (0.64 to 0.85)	0.74 (0.64 to 0.85)	0.45 (-0.11 to 1.01)	0.84 (0.59 to 1.19)	
< -3	1.56 (1.25 to 1.95)	0.97 (0.74 to 1.27)	0.62 (0.45 to 0.86)			

Table 3. Sensitivity analysis using Cox proportional hazards regression to evaluate the association between biannual oral azithromycin distribution and mortality by WAZ subgroups<sup>1</sup>.

<sup>1</sup>As treatment arm was randomized and is the primary intervention of interest, confounding of the relationship between nutritional status and mortality was not considered, and no additional factors were controlled for in the model.

<sup>2</sup>For this calculation, subgroups were coded so the groups with the lowest mortality rates (azithromycin arm, higher WAZ subgroup) were the reference categories. **Abbreviations:** HR, hazard ratio; RERI<sub>HR</sub>, Relative Excess Risk due to Interaction from hazard ratios, WAZ, weight-for-age Z-score.

https://doi.org/10.1371/journal.pmed.1003285.t003

In summary, a placebo-controlled trial found that biannual azithromycin distribution reduced mortality among children 1–11 months old regardless of underweight status. Although the observed mortality reduction with azithromycin was larger among subgroups of underweight children, underweight status was not a statistically significant effect modifier in this trial. Treatment of all children 1–11 months old would save 5 times as many lives as restricting treatments only to children with a WAZ < -3.

# **Supporting information**

**S1 CONSORT Checklist. Details about where the trial-specific information outlined by the CONSORT guidelines can be found in this manuscript.** CONSORT, Consolidated Standards of Reporting Trials. (DOC)

S1 Table. Summary of characteristics of children 1–11 months old at the time of entry into the study among included children (n = 27,222) and excluded children (n = 12,086). Excluded children include those who were 1–11 months of age at the time of entry into the study and did not have weight measured (n = 11,899) or had an invalid weight recorded (n = 187). (DOCX)

# **Author Contributions**

**Conceptualization:** Kieran S. O'Brien, Robin L. Bailey, Sheila K. West, Travis C. Porco, Benjamin Arnold, Jeremy D. Keenan, Thomas M. Lietman.

- **Data curation:** Kieran S. O'Brien, Robin L. Bailey, Sheila K. West, Travis C. Porco, Jeremy D. Keenan, Thomas M. Lietman.
- Formal analysis: Kieran S. O'Brien.
- Funding acquisition: Robin L. Bailey, Sheila K. West, Jeremy D. Keenan, Thomas M. Lietman.

**Investigation:** Kieran S. O'Brien, Ahmed M. Arzika, Robin L. Bailey, Sheila K. West, Catherine E. Oldenburg, Travis C. Porco, Benjamin Arnold, Jeremy D. Keenan, Thomas M. Lietman.

- Methodology: Kieran S. O'Brien, Robin L. Bailey, Sheila K. West, Catherine E. Oldenburg, Travis C. Porco, Benjamin Arnold, Jeremy D. Keenan, Thomas M. Lietman.
- Project administration: Kieran S. O'Brien, Ahmed M. Arzika, Ramatou Maliki, Farouk Manzo, Alio K. Mamkara, Elodie Lebas, Catherine Cook, Robin L. Bailey, Sheila K. West, Jeremy D. Keenan, Thomas M. Lietman.
- Supervision: Ahmed M. Arzika, Ramatou Maliki, Farouk Manzo, Alio K. Mamkara, Elodie Lebas, Catherine Cook, Robin L. Bailey, Sheila K. West, Catherine E. Oldenburg, Jeremy D. Keenan, Thomas M. Lietman.
- Visualization: Kieran S. O'Brien.
- Writing original draft: Kieran S. O'Brien.
- Writing review & editing: Kieran S. O'Brien, Ahmed M. Arzika, Ramatou Maliki, Farouk Manzo, Alio K. Mamkara, Elodie Lebas, Catherine Cook, Robin L. Bailey, Sheila K. West, Catherine E. Oldenburg, Travis C. Porco, Benjamin Arnold, Jeremy D. Keenan, Thomas M. Lietman.

#### References

- Keenan JD, Bailey RL, West SK, Arzika AM, Hart J, Weaver J, et al. Azithromycin to reduce childhood mortality in sub-Saharan Africa. The New England journal of medicine. 2018; 378(17):1583–92. Epub 2018/04/26. https://doi.org/10.1056/NEJMoa1715474 PMID: 29694816; PubMed Central PMCID: PMC5849140.
- Keenan JD, Arzika AM, Maliki R, Boubacar N, Elh Adamou S, Moussa Ali M, et al. Longer-Term Assessment of Azithromycin for Reducing Childhood Mortality in Africa. The New England journal of medicine. 2019; 380(23):2207–14. Epub 2019/06/06. https://doi.org/10.1056/NEJMoa1817213 PMID: 31167050; PubMed Central PMCID: PMC6512890.
- Doan T, Arzika AM, Hinterwirth A, Maliki R, Zhong L, Cummings S, et al. Macrolide Resistance in MOR-DOR I—A Cluster-Randomized Trial in Niger. The New England journal of medicine. 2019; 380 (23):2271–3. Epub 2019/06/06. https://doi.org/10.1056/NEJMc1901535 PMID: 31167060; PubMed Central PMCID: PMC6518950.
- O'Brien KS, Emerson P, Hooper PJ, Reingold AL, Dennis EG, Keenan JD, et al. Antimicrobial resistance following mass azithromycin distribution for trachoma: a systematic review. The Lancet Infectious diseases. 2019; 19(1):e14–e25. Epub 2018/10/08. https://doi.org/10.1016/S1473-3099(18)30444-4 PMID: 30292480.
- Austin DJ, Kristinsson KG, Anderson RM. The relationship between the volume of antimicrobial consumption in human communities and the frequency of resistance. Proc Natl Acad Sci U S A. 1999; 96 (3):1152–6. Epub 1999/02/03. <u>https://doi.org/10.1073/pnas.96.3.1152</u> PMID: 9927709; PubMed Central PMCID: PMC15366.
- Black RE, Victora CG, Walker SP, Bhutta ZA, Christian P, de Onis M, et al. Maternal and child undernutrition and overweight in low-income and middle-income countries. The Lancet. 2013; 382(9890):427– 51. https://doi.org/10.1016/S0140-6736(13)60937-X
- Scrimshaw NS, SanGiovanni JP. Synergism of nutrition, infection, and immunity: an overview. The American journal of clinical nutrition. 1997; 66(2):464S–77S. Epub 1997/08/01. https://doi.org/10.1093/ ajcn/66.2.464S PMID: 9250134.
- Dewey KG, Mayers DR. Early child growth: how do nutrition and infection interact? Maternal & child nutrition. 2011; 7 Suppl 3:129–42. Epub 2011/10/05. https://doi.org/10.1111/j.1740-8709.2011.00357.x PMID: 21929641
- Langtry HD, Balfour JA. Azithromycin. A review of its use in paediatric infectious diseases. Drugs. 1998; 56(2):273–97. Epub 1998/08/26. https://doi.org/10.2165/00003495-199856020-00014 PMID: 9711451.
- Humphrey JH. Child undernutrition, tropical enteropathy, toilets, and handwashing. Lancet (London, England). 2009; 374(9694):1032–5. Epub 2009/09/22. https://doi.org/10.1016/s0140-6736(09)60950-8 PMID: 19766883.
- Lunn PG. The impact of infection and nutrition on gut function and growth in childhood. Proceedings of the Nutrition Society. 2007; 59(1):147–54. Epub 02/28. https://doi.org/10.1017/S0029665100000173 PMID: 10828184

- Berkley JA, Ngari M, Thitiri J, Mwalekwa L, Timbwa M, Hamid F, et al. Daily co-trimoxazole prophylaxis to prevent mortality in children with complicated severe acute malnutrition: a multicentre, double-blind, randomised placebo-controlled trial. The Lancet Global health. 2016; 4(7):e464–73. Epub 2016/06/07. https://doi.org/10.1016/S2214-109X(16)30096-1 PMID: 27265353.
- 13. Rafii F, Sutherland JB, Cerniglia CE. Effects of treatment with antimicrobial agents on the human colonic microflora. Therapeutics and clinical risk management. 2008; 4(6):1343–58. Epub 2009/04/02. https://doi.org/10.2147/tcrm.s4328 PMID: 19337440; PubMed Central PMCID: PMC2643114.
- Sullivan A, Edlund C, Nord CE. Effect of antimicrobial agents on the ecological balance of human microflora. The Lancet Infectious diseases. 2001; 1(2):101–14. Epub 2002/03/02. <u>https://doi.org/10.1016/</u> S1473-3099(01)00066-4 PMID: 11871461.
- Trehan I, Goldbach HS, LaGrone LN, Meuli GJ, Wang RJ, Maleta KM, et al. Antibiotics as part of the management of severe acute malnutrition. The New England journal of medicine. 2013; 368(5):425–35. Epub 2013/02/01. https://doi.org/10.1056/NEJMoa1202851 PMID: 23363496; PubMed Central PMCID: PMC3654668.
- Isanaka S, Langendorf C, Berthe F, Gnegne S, Li N, Ousmane N, et al. Routine Amoxicillin for Uncomplicated Severe Acute Malnutrition in Children. The New England journal of medicine. 2016; 374 (5):444–53. Epub 2016/02/04. https://doi.org/10.1056/NEJMoa1507024 PMID: 26840134.
- Vray M, Hedible BG, Adam P, Tondeur L, Manirazika A, Randremanana R, et al. A multicenter, randomized controlled comparison of three renutrition strategies for the management of moderate acute malnutrition among children aged from 6 to 24 months (the MALINEA project). Trials. 2018; 19(1):666. Epub 2018/12/06. https://doi.org/10.1186/s13063-018-3027-3 PMID: <u>30514364</u>; PubMed Central PMCID: PMC6278112.
- Oldenburg CE, Arzika AM, Maliki R, Kane MS, Lebas E, Ray KJ, et al. Safety of azithromycin in infants under six months of age in Niger: A community randomized trial. PLoS Negl Trop Dis. 2018; 12(11): e0006950. Epub 2018/11/13. https://doi.org/10.1371/journal.pntd.0006950 PMID: 30419040; PubMed Central PMCID: PMC6258425.
- WHO Multicentre Growth Reference Study Group. Assessment of differences in linear growth among populations in the WHO Multicentre Growth Reference Study. Acta paediatrica (Oslo, Norway: 1992) Supplement. 2006; 450:56–65. Epub 2006/07/05. <u>https://doi.org/10.1111/j.1651-2227.2006.tb02376.x</u> PMID: 16817679.
- 20. WHO Multicentre Growth Reference Study Group. WHO Child Growth Standards: Length/height-forage, weight-for-age, weight-for-length, weight-for-height and body mass indexfor-age: Methods and development. Geneva: World Health Organization, 2006. [cited 2020 Aug 21]. https://www.who.int/ childgrowth/standards/technical\_report/en/
- R Core Team. R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing; 2003. http://www.R-project.org/.
- VanderWeele TJ. Sample size and power calculations for additive interactions. Epidemiologic Methods. 2012; 1:159–88. <u>https://doi.org/10.1515/2161-962X.1010</u> PubMed Central PMCID: PMC4249707. PMID: 25473594
- VanderWeele TJ, Knol MJ. A tutorial on interaction. Epidemiologic Methods. 2014; 3:33–72. [cited 2020 Aug 21]. https://cdn1.sph.harvard.edu/wp-content/uploads/sites/603/2018/04/InteractionTutorial\_EM. pdf
- Knol MJ, VanderWeele TJ, Groenwold RH, Klungel OH, Rovers MM, Grobbee DE. Estimating measures of interaction on an additive scale for preventive exposures. Eur J Epidemiol. 2011; 26(6):433–8. Epub 2011/02/24. https://doi.org/10.1007/s10654-011-9554-9 PMID: 21344323; PubMed Central PMCID: PMC3115067.
- Knol MJ, VanderWeele TJ. Recommendations for presenting analyses of effect modification and interaction. International journal of epidemiology. 2012; 41(2):514–20. Epub 2012/01/19. <u>https://doi.org/10. 1093/ije/dyr218</u> PMID: 22253321; PubMed Central PMCID: PMC3324457.
- Li R, Chambless L. Test for additive interaction in proportional hazards models. Ann Epidemiol. 2007; 17(3):227–36. Epub 2007/02/27. https://doi.org/10.1016/j.annepidem.2006.10.009 PMID: 17320789.
- **27.** Pelletier DL. The relationship between child anthropometry and mortality in developing countries: implications for policy, programs and future research. The Journal of nutrition. 1994; 124(10 Suppl):2047s–81s. Epub 1994/10/01. https://doi.org/10.1093/jn/124.suppl\_10.2047S PMID: 7931716.
- McDonald CM, Olofin I, Flaxman S, Fawzi WW, Spiegelman D, Caulfield LE, et al. The effect of multiple anthropometric deficits on child mortality: meta-analysis of individual data in 10 prospective studies from developing countries. The American journal of clinical nutrition. 2013; 97(4):896–901. Epub 2013/ 02/22. https://doi.org/10.3945/ajcn.112.047639 PMID: 23426036.
- Tam CC, Offeddu V, Lim JM, Voo TC. One drug to treat them all: ethical implications of the MORDOR trial of mass antibiotic administration to reduce child mortality. J Glob Health. 2019; 9(1):010305. Epub

2019/01/16. https://doi.org/10.7189/jogh.09.010305 PMID: 30643634; PubMed Central PMCID: PMC6318831.

- Brander RL, Weaver MR, Pavlinac PB, John-Stewart GC, Hawes SE, Walson JL. Projected impact and cost-effectiveness of community-based versus targeted azithromycin administration strategies for reducing child mortality in sub-Saharan Africa. Clinical infectious diseases: an official publication of the Infectious Diseases Society of America. 2020. Epub 2020/01/07. <u>https://doi.org/10.1093/cid/ciz1220</u> PMID: 31905386.
- **31.** Rose G. Sick individuals and sick populations. International journal of epidemiology. 1985; 14(1):32–8. Epub 1985/03/01. https://doi.org/10.1093/ije/14.1.32 PMID: 3872850.
- Chowkwanyun M, Bayer R, Galea S. "Precision" Public Health—Between Novelty and Hype. New England Journal of Medicine. 2018; 379(15):1398–400. https://doi.org/10.1056/NEJMp1806634 PMID: 30184442.
- **33.** Oldenburg CE, Arzika AM, Maliki R, Lin Y, O'Brien KS, Keenan JD, et al. Optimizing the Number of Child Deaths Averted with Mass Azithromycin Distribution. The American journal of tropical medicine and hygiene. 2020. Epub 2020/02/19. https://doi.org/10.4269/ajtmh.19-0328 PMID: 32067626.
- Akombi BJ, Agho KE, Merom D, Renzaho AM, Hall JJ. Child malnutrition in sub-Saharan Africa: A meta-analysis of demographic and health surveys (2006–2016). PLoS ONE. 2017; 12(5):e0177338. Epub 2017/05/12. https://doi.org/10.1371/journal.pone.0177338 PMID: <u>28494007</u>; PubMed Central PMCID: PMC5426674.
- 35. O'Brien KS, Amza A, Kadri B, Nassirou B, Cotter SY, Stoller NE, et al. Comparison of anthropometric indicators to predict mortality in a population-based prospective study of children under 5 years in Niger. Public health nutrition. 2020; 23(3):538–43. Epub 2019/09/10. https://doi.org/10.1017/ S1368980019002520 PMID: 31496465; PubMed Central PMCID: PMC7024038.
- Myatt M, Khara T, Dolan C, Garenne M, Briend A. Improving screening for malnourished children at high risk of death: a study of children aged 6–59 months in rural Senegal. Public health nutrition. 2019; 22(5):862–71. Epub 2018/12/07. https://doi.org/10.1017/S136898001800318X PMID: 30501655; PubMed Central PMCID: PMC6521791.
- 37. Myatt M, Khara T, Schoenbuchner S, Pietzsch S, Dolan C, Lelijveld N, et al. Children who are both wasted and stunted are also underweight and have a high risk of death: a descriptive epidemiology of multiple anthropometric deficits using data from 51 countries. Archives of public health = Archives belges de sante publique. 2018; 76:28. Epub 2018/07/22. https://doi.org/10.1186/s13690-018-0277-1 PMID: 30026945; PubMed Central PMCID: PMC6047117.
- Vaitla B, Devereux S, Swan SH. Seasonal hunger: a neglected problem with proven solutions. PLoS Med. 2009; 6(6):e1000101. Epub 2009/07/01. https://doi.org/10.1371/journal.pmed.1000101 PMID: 19564900; PubMed Central PMCID: PMC2696035.
- 39. Porco TC, Hart J, Arzika AM, Weaver J, Kalua K, Mrango Z, et al. Mass Oral Azithromycin for Childhood Mortality: Timing of Death After Distribution in the MORDOR Trial. Clinical infectious diseases: an official publication of the Infectious Diseases Society of America. 2019; 68(12):2114–6. Epub 2018/12/19. https://doi.org/10.1093/cid/ciy973 PMID: 30561577; PubMed Central PMCID: PMC6541729.
- Perumal N, Bassani DG, Roth DE. Use and Misuse of Stunting as a Measure of Child Health. The Journal of nutrition. 2018; 148(3):311–5. Epub 2018/03/17. <u>https://doi.org/10.1093/jn/nxx064</u> PMID: 29546307.